



Article Identification and Reduction of Product Carbon Footprints: Case Studies from the Austrian Automotive Supplier Industry

Kai Rüdele 🗅 and Matthias Wolf *🕩

Institute of Innovation and Industrial Management, Graz University of Technology, 8010 Graz, Austria; kai.ruedele@tugraz.at

* Correspondence: matthias.wolf@tugraz.at; Tel.: +43-316-873-7796

Abstract: Greenhouse gas (GHG) emissions from human activities have climbed significantly above pre-pandemic levels and have reached record highs that unequivocally accelerate global warming. Industry has a significant impact on climate change, emitting at least 21% of global GHGs and making little overall progress toward its reduction until now. Reducing industry's emissions requires coordinated action along the value chains in order to promote mitigation options, such as energy and material efficiency, circular material flows, and transformative changes within production processes. This article is the first evaluation of GHG emissions generated during the manufacturing of vehicle components by Austrian companies. For this, the authors analyzed three different products of automotive suppliers according to the methodology of ISO 14067. Despite previous efforts toward an environmentally compatible fabrication, additional and significant reduction potentials were identified. These measures for product carbon footprint (PCF) reduction included the sourcing of lowcarbon materials (which are already available on the market), more extensive use of renewable energy, and changes towards more resource efficient manufacturing processes and machinery. Depending on the materials used, the PCF can be reduced by up to 80%. The findings serve to prepare for future PCF reporting regulations and illustrate reduction potentials to achieve future market advantages, especially when PCFs become an awarding criterion.

Keywords: carbon footprint; decarbonization; car manufacturing; automotive industry; low-carbon material

1. Introduction

Global GHG emissions are back above pre-pandemic levels. According to the International Monetary Fund [1], manufacturing has been a particularly high contributor to recent global increases. In the meantime, energy use from industry contributes to almost a quarter of global GHG emissions, with an additional 5% coming from direct industrial processes (not including any emissions caused by freight transport) [2]. This trend, which has persisted for at least three decades, is all the more alarming as we know that climate change and global warming are mainly caused by anthropogenic emissions [3,4].

Data provided by the OECD reveal that GHG emissions from manufacturing industries and construction in developed countries have declined significantly [5]. Despite considerable economic growth and gains in prosperity, the seven largest advanced economies in the world (G7) reduced their emissions in this segment by 22% between 1990 and 2021; the member states of the European Union (EU27) were able to reduce this amount by 39% during the same period. However, these figures should be treated with skepticism as this is partly caused by a shift in the trade structure of high-income countries, i.e., domestically produced goods being substituted by imported ones [6]. As a result, emerging countries, especially China and India, have—despite their reduction of territorial emissions—multiplied their GHG emissions over the past thirty years because of increased net imports and exports of emissions. This shift in trade and the subsequent relocation of emissions from imported



Citation: Rüdele, K.; Wolf, M. Identification and Reduction of Product Carbon Footprints: Case Studies from the Austrian Automotive Supplier Industry. *Sustainability* **2023**, *15*, 14911. https://doi.org/10.3390/su152014911

Academic Editors: Sheila Devasahayam and Sunil Kumar Tripathy

Received: 11 September 2023 Revised: 6 October 2023 Accepted: 10 October 2023 Published: 16 October 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). goods creates the illusion that developed countries only need to reduce their territorial emissions. Attributing embedded emissions of goods to the exporting producer rather than the importing consumer (country) affects priorities and mitigation possibilities [7]. Existing and upcoming carbon tariffs such as the Carbon Border Adjustment Mechanism (CBAM) [8] will counteract this misallocation (known as 'carbon leakage') and lead to a more holistic view. Possible consequences could include changes to sourcing, up to local in-house production; the latter would lower transportation burdens and the associated emissions [6]. However, contrary effects, including an increase in global GHG emissions, may also occur [9].

Contrasting these developments, the Intergovernmental Panel on Climate Change (IPCC) is urging immediate action to tackle climate change by reducing GHG emissions and limiting global warming to 1.5 °C to 2 °C above pre-industrial levels, thus avoiding severe and irreversible destruction of the basis of human existence. Its latest assessment report emphasizes the need for deploying innovative processes and practices to enable industry's transition to net zero emissions [10]. According to the report, the most effective measures to decarbonize production in the medium term include implementing low-emission electricity and heat, energy efficiency improvements via best available technologies, fuel switching, and the substitution of high-carbon feedstock. Moreover, ecodesign, material efficiency, and waste reduction, as well as higher recycling rates up to circular economy, are the drivers for deep decarbonization trajectories [10].

Various initiatives have been launched to reduce GHG emissions, both at the national and international levels. The most ambitious set of policy initiatives comes from the European Commission's so-called Green Deal, with its declared goal of making Europe the first climate-neutral continent by 2050. This goal is legally anchored in the European Climate Law, which also sets the intermediate target of reducing net greenhouse gas emissions by at least 55% by 2030 compared with 1990 levels [11]. One featured action package is the Green Deal Industrial Plan, focusing on a simplified regulatory environment, faster access to funding, and green jobs [12]. This plan supplements previous actions of the European Union, such as the Energy Efficiency Directive 2012, the Renewable Energy Directive 2018, the Emissions Trading System (EU-ETS), and the EU taxonomy for sustainable activities. The recent report on the functioning of the European carbon market shows that in 2021, around EUR 19.4 billion (76% of total ETS auction revenues) was used for climate- and energy-related projects [13]. These investments into renewable energy, energy efficiency, and research may help the member states move closer to a net-zero emissions pathway.

Individual countries have announced their intentions to become carbon neutral within the next few decades [14,15]. For instance, Austria's federal government is pursuing the target of completely decarbonizing its energy sector and the whole of its economic system by 2040. The legislative program released in January of 2020 adds to the climate ambitions previously adopted as part of its 2018 climate and energy strategy. The new government brought the carbon neutrality target forward by a decade from 2050 to 2040 [16]. This results in the ambitious decarbonization paths that are illustrated in Figure 1.



Figure 1. Austria's territorial annual GHG emissions and decarbonization paths (based on [5,17]).

Since peaking in 2005, Austria's total GHG emissions have declined steadily. In 2020, due to the Corona pandemic, less than 75 million tons of CO₂ equivalents (CO₂-eq) were emitted, but they started to rise again thereafter. The decrease over several years reflects the effectiveness of climate protection measures taken at that time. Occasional increases can be traced back to low prices for fossil energy and strong economic development [18]. However, considering the minimal reduction in emissions observed since 2005, a much faster transformation of the industrial sector is required in order to operate without fossil energy by 2040, thus contributing to the goal of climate neutrality. Aligned to this, an evaluation by the European Commission indicates that Austria's aimed targets for energy efficiency (especially regarding primary and final energy consumption) are lacking in ambition [19]. To achieve the 2040 target of net-zero emissions without exceeding the Austrian GHG budget of the industrial sector (see Figure 1), it is necessary that the demand for electricity, district heating, and industrial goods is reduced through behavioral changes (e.g., sharing economy, sufficiency) and measures that increase energy and resource efficiency. Additionally, the energy system must be converted to renewable energy sources [18].

A related study by Diendorfer et al. [20] determined the technical decarbonization potentials of electrification, the use of carbon-neutral gas, circular economy, and carbon capture for several sectors of Austrian industry. In the best-case scenario, these decarbonization options taken together yield an annual saving of 22 Mt CO_2 -eq. The total investment costs associated with the implementation of these options up to the year 2040 depend on the chosen technologies and range from EUR 6.2 billion to 11.2 billion (excluding operating costs). The study focuses on energy-intensive industrial sectors (namely iron and steel production, the chemical industry, the paper industry, and the stone and glass sector) and does not examine emission drivers and approaches to cut GHG emissions for other sectors. Avoiding emissions through the combustion of fossil fuels is the central point, whereas the reduction of embodied carbon emissions by material efficiency and substitution is subsidiary.

Furthermore, the issue of sustainability is gathering momentum in the automotive industry. Along with both a shortage of resources and the standards and regulations put in place to penalize pollution, heightened environmental awareness on the part of customers may also push the automotive industry toward more ecological production [21]. The potential is available in that GHG emissions from car production of over 7 kg of CO₂-eq per kg of vehicle weight can be more than halved through renewable energy and sustainable materials [22].

Regarding the aforementioned recent developments, a rigorous change is required that will focus on the product's life cycle, including all embodied emissions, irrespective of their origin. A life cycle assessment (LCA) of product–production systems helps to identify potentials to reduce environmental impacts and evaluate viable solutions and alternatives. In engineering science, the LCA has become established not only as a sustainability assessment tool but also as a research methodology [23,24]. Like other quantitative approaches, an LCA can assist or improve a decision-making process and is beneficial for operational research [25,26].

The carbon footprint is a study based on a broader LCA and expresses the quantity of GHGs (measured in kg CO_2 -eq) emitted into the atmosphere by an individual, organization, process, product, or event from within a specified boundary [27]. If the scope of the study starts with the extraction of raw materials ('Cradle-to-x') and considers all upstream activities, the embedded emissions of imported goods become transparent.

This paper presents the authors' approach to determine the product carbon footprint (PCF) of three components produced by Austrian automotive suppliers: a crossmember, a rear axle carrier, and a grill shutter actuator. By compiling measured and calculated inputs and outputs, data were generated from which the amount of released GHG was determined. To our knowledge, this study is the first of its kind considering industrial goods manufactured in Austria alongside associated characteristics such as electricity mix, supply chains, and process maturity. It reports the main challenges and limitations of data collection and analysis but also provides recommendations for a structured procedure, since there are more than 60 initiatives and methods for calculating and reporting PCFs [28]. The results are used to identify the largest emission drivers and to quantify potential savings.

This article is structured as follows: Section 2 describes the main characteristics of the (Austrian) automotive industry, primarily with regard to economic and environmental aspects. Section 3 summarizes and reviews relevant PCF studies in the automotive sector, while Section 4 presents and discusses the empirical findings from Austrian automotive suppliers. Finally, Section 5 includes the conclusion and suggestions for further research.

2. Characteristics of the Automotive Industry

2.1. Economic Relevance

With regard to revenue and resource consumption, the automotive industry is one of the world's most important economic sectors. In the last five years, the world's annual motor vehicle production, comprising both passenger cars and commercial vehicles, ranged between 78 and 96 million units [29]. More detailed insights on the six largest vehicle producing countries, as well as Austria, are provided in Table 1. The market volume of global automobile production amounts to USD 3 trillion per year [30] and is expected to reach USD 3.8 trillion by 2030 [31]. The European automotive sector employs 13 million people directly and indirectly, accounting for 7% of all EU jobs and 11.5% of all EU manufacturing jobs [32].

Country	Passenger Cars	Commercial Vehicles	Total Production	Change Compared with 2021	Share of Global Production
Austria	108 (125)	0 (12)	108 (137)	-21%	0.1%
China	23,836 (21,408)	3185 (4674)	27,021 (26,082)	3%	31.8%
Germany	3480 (3096)	197 (213)	3678 (3309)	11%	4.3%
India	4439 (3631)	1018 (768)	5457 (4399)	24%	6.4%
Japan	6566 (6619)	1269 (1228)	7836 (7847)	0%	9.2%
South Korea	3438 (3163)	319 (300)	3757 (3462)	9%	4.4%
United States	1752 (1563)	8309 (7604)	10,060 (9167)	10%	11.8%
World	61,599 (57,054)	23,418 (23,092)	85,017 (80,146)	6%	

Table 1. Production statistics from 2022 in 1000 units, with the previous year's figures in brackets [29].

Although not as substantial as in other European countries [33,34], the automotive (component) industry has long been an important pillar of the Austrian economy. According to Statistik Austria [35], it directly employs at least 37,000 people and accounts for over

8% of the total revenue and 6% of the gross value added (GVA) of the total manufacturing sector (ISIC section C). In 2022, the GVA of the Austrian automotive industry (C 29 ISIC) amounted to USD 3.1 billion, which was equal to one percent of the national GVA [36]. In addition to the production of passenger cars, motor vehicle components (1.6 million engines and transmissions in 2021) and motorcycles (179,000 in 2021) are also manufactured in Austria [37].

2.2. Current Environmental Impact of Automobile Production

Even though Austrian automotive companies only emit a fraction (154,000 t CO_2 -eq) of what other industries (e.g., manufacturing of machinery: 430,000 t CO_2 -eq, [38]) emit, businesses in the transportation sector are particularly challenged to reduce their emissions, as road vehicles alone account for 16% of global energy-related CO_2 emissions [39]. New technologies and services, regulations, and usage patterns of vehicles have already transformed or even disrupted the auto industry [40]. Consequently, environmentally sound production processes represent an additional challenge for this industry.

By definition, the automotive sector does not belong to the energy-intensive industries [41–43]; however, its sourcing activities in those industries are significant. The European Steel Association estimates that the automotive sector accounts for 17% of steel demand in the EU, especially strip mill products [44]. On average, 900 kg of steel is used per passenger car, mainly in the body structure, drive train, panels, doors, and suspension [45]. With a share of 18%, the manufacture of vehicles is also one of the main industrial consumers of aluminum globally [46]. In 2022, the total aluminum gross demand for the European automotive industry amounted to more than 2.8 million tons, of which nearly half was processed through energy-intensive casting [47]. With a production volume of 11 million tons in 2022, the flat glass sector represents the second largest sector of the European glass industry [48], of which almost 15% is processed into glazing for the automotive and transport industry [49]. As Sato and Nakata [50] demonstrate, a large quantity of copper ore is also needed for vehicle production. About 6 to 18% of a vehicle's curb weight is made up of plastic, whereas rubber contributes to approximately 4 to 7% [50–53]. Due to their tendency to have a lower global warming potential, (synthetic) polymers are less relevant than metals (e.g., [54,55]).

Damert and Baumgartner [56] found that final vehicle manufacturers (original equipment manufacturers (OEMs)) are more ambitious in carbon compensation than their suppliers due to their greater exposure to public attention and associated stakeholder pressure. Because of their size, they also have better access to financial resources and knowledge. While OEM companies often come under scrutiny because they are the main party held responsible for tailpipe and fleet emissions, car part manufacturers play a key role in cutting emissions caused by manufacturing. Currently, there are few first-tier suppliers delivering complete systems and modules, whereby they expanded their capabilities, dominate the technologies of a respective sector, and contribute to innovation significantly [57]. However, the changeover to low-carbon products and production was not rewarding until recently due to a lack of demand and high investments [58]. Nevertheless, there are numerous concepts to reduce supply chain emissions by involving (sub)suppliers and multiple tiers, such as low-carbon procurement or product stewardships (e.g., [59,60]). Already, more than twelve years ago, Lee [61] demonstrated how OEMs could work with their suppliers to improve the environmental performance of manufacturing processes by measuring the PCF of single components. An index proposed by Azevedo and Barros [62] showed that (environmental) sustainability of the UK's automotive supply chain has improved in recent years.

There are also characteristics of the automotive industry that come with positive effects for the environment. On the one hand, it is operating under considerable cost pressure, and the predominate sector adopting lean principles results in an efficient resource consumption [63,64]. On the other hand, the automotive industry is characterized by high volumes and automation, resulting in corresponding economies of scale. One-off

productions and their associated resource-intensive and polluting manufacturing processes, such as lost-wax casting [65], are the absolute exception.

2.3. Trends and Challenges in the Automotive Sector

Forecasts predict that vehicle production and sales will remain above current levels (at least until 2026) and that SUVs will remain the most popular segment [66]. Vehicles are also becoming larger and heavier (e.g., [67]). Estimates assumed that this resulted in the additional consumption of up to 9.2 million tons of steel in 2022 [68]. With the mass market transition to electric vehicles and the accompanying lighter-weight construction, the average aluminum content per car manufactured in Europe is expected to increase from today's 205 kg to 256 kg by 2030 [47]. By 2050, the automotive industry's demand for aluminum is even predicted to double [46]. But each kilogram of aluminum that substitutes steel in automotive applications saves up to 20 kg of CO₂-eq over the life of the vehicle [69]. In the future, emissions from production will have a higher share than those from the entire use phase. According to the World Economic Forum [53], 60% of automotive life cycle emissions will come from materials by 2040. Both aluminum and steel are highly recyclable, however there are partly contradictory statements regarding the extent to which this potential is exploited. While the European Aluminium Association [70] states recycling rates reach over 90% in the automotive sector, the International Aluminium Institute [71] forecasts that, even by 2050, primary aluminum will still constitute approximately 45% of the aluminum used by the automobile industry. According to the American Iron and Steel Institute [72], the steel industry is recycling nearly 100% of the steel in automobiles by the end of their life cycles. A large number of studies encourage vehicle production to intensify the use of recyclates, selective disassembly to single out hazardous or valuable components, and separation processes to liberate desired materials to save energy, costs, and emissions (e.g., [51,73–75]). Greenpeace demands a transition to zero-carbon steel [68].

2.4. Actions of the Automotive Industry

On their websites, nearly all vehicle manufacturers publish extensive sustainability reports that examine not only current and future fleet consumption but also the environmental burden of vehicle production. In addition to actual performance indicators, the reports also include targets and measures for decarbonization and resource conservation. For European car manufacturers, such reporting has already been firmly established for more than 10 years [76], and Lukin et al. [77] cursorily examined the extent to which leading automotive companies meet certain Sustainable Development Goals (SDG) of the United Nations. Meanwhile, most sustainability reports are following standards provided by the Global Reporting Initiative (GRI) [78]. A GRI compliant sustainability report must contain a section focusing on used materials and supply chain-related sustainability issues (especially 'procurement practices' and 'supplier environmental assessment') [79]. An analysis by Caliskan et al. [80] suggested that sustainability reports from automotive suppliers tended to have a weaker stance against SDGs than those of OEMs.

OEMs use these reports to announce their own milestones. BMW [81], for example, is striving for the 'lowest possible level of resource consumption in production', while Toyota [82] wants to 'reduce CO₂ emissions from global plants by 30% compared to 2013 levels' and 'achieve carbon neutrality at all global plants by 2035'. Such ambitious plans present competitive opportunities for suppliers to aid OEMs with their targets. An overview regarding low-carbon automotive steel partnerships is provided by Liu et al. [68].

OEMs are perceived as more (pro)active than suppliers when it comes to implementing climate change measures [56] as they focus more on carbon compensation measures and engagement with policymakers to legitimize their business operations. Damert and Baumgartner [56] also recommend that OEMs should use their buying power to encourage their suppliers to take more ambitious actions on climate change.

The initiation of low-carbon products can start with carbon-accounting and monitoring activities that evaluate the GHG emissions caused by suppliers (e.g., [83]). As long as

20 years ago, the German Association of the Automotive Industry (VDA) introduced the first version of its LCA Data Collection Format as a means for collecting, processing, and documenting relevant data to enable automobile manufacturers and their partners along the supply chain to environmentally improve their products. The related documents shall be applied in order to use the obtained data to evaluate (design) options and justify environment-related decisions concerning the production of automotive components and operating materials for automobiles, as well as processes used for production, recovery, and disposal. The main purpose is to simplify and standardize the compilation and quantification of inputs and outputs for a product (life cycle inventory, LCI) for both data suppliers and users [84,85].

3. Previous Studies to Determine Carbon Footprints in the Automotive Supply Industry

All studies known to the authors show that the upstream carbon footprint accounts for a substantial proportion (at least 75%) of the overall climate impact of motor vehicle manufacturing (e.g., [86,87]). This coincides with the sustainability reports from OEMs. The GHG emissions from the production of suppliers (Scope 3 category 'purchased goods' according to GHG Protocol, [88]) surpass those of manufacturing and the final assembly by an OEM by several times. Table 2 compares the information of eight of the ten OEMs with the highest number of sales according to Forbes [89]. The two missing companies either did not publish a suitable report (SAIC Motor) or the data did not allow for a breakdown of the automotive business (Honda Motor). Unless otherwise stated, the values related to all brands of the respective group. If no data per produced vehicle were reported, the total emissions of the particular scope were divided by the production volume. Low values for 'Logistics' suggest that only upstream or downstream transportation and distribution were reported.

Manufacturer (Group)	Supply Chain (Purchased Goods)	Production by OEM	Logistics	Source
BMW	8.3	0.3	1.0	[90]
Ford	9.5	0.7	1.8	[91]
General Motors	8.1	0.7	2.4	[92]
Hyundai	5.0	0.6	0.3	[93]
Mercedes-Benz Group	8.7	0.3	1.1	[94]
Stellantis	6.7	0.6	0.2	[95]
Toyota	10.5	0.8	0.6	[82]
Volkswagen	9.3	0.7	0.5	[96]

Table 2. 2022 GHG emissions in t CO_2 -eq per vehicle.

Annotations: Hyundai w/o Kia; Toyota data from 2021 and w/o Daihatsu; Volkswagen limited to passenger cars and light commercial vehicles of all brands.

In addition to OEMs and suppliers' own disclosures, there are numerous publications in the field of LCA, both at the overall vehicle level and for individual components. The analyses of Sullivan et al. [52] and Sato and Nakata [50] used data from numerous sources to estimate material composition, determine involved processes, and calculate the cumulative energy consumption of car production. While Sato and Nakata [50] only detailed resource consumption and did not consider the environmental burdens, Sullivan et al.'s work pointed out related CO₂ emissions (but not the total GHG-equivalent emissions) [52]. Gebler et al. [97] not only compared already known production-related GHG emissions but also analyzed the relative influence of annual production volume on the resource demands of an automotive factory. The findings appeared to be consistent with Sullivan et al., wherein about 40% of the energy consumption and carbon emissions for part manufacturing and vehicle assembly were fixed and resulted from the plant's base load and production readiness, which included lighting, heat, ventilation, air conditioning (HVAC), and material handling [52]. Rivera and Reyes-Carrillo [98] had already shown that over 80% of the hazardous emissions from automobile manufacturing were associated with the painting stage, and their work also provided guidance to assess potential solutions. Finally, Kim and Wallington [99] discovered that the materials used in lightweight vehicles were more carbon intensive to extract and manufacture in many cases.

Nearly the same applies for suppliers and vendor parts: in their case study, Cecchel et al. [100] assessed a suspension crossbeam for commercial vehicles, concluding that the extraction of primary aluminum was by far the most energy-intensive stage in the supply chain, followed by die casting. The situation was similar in a case study that calculated the energy demand and GHG emissions of front subframes, which showed that the material production predominated the parts manufacturing in all scenarios [101]. For a rear crash management system, Del Pero [102,103] revealed that, independent of the selected metal, the raw material extraction and production caused more GHG emissions than the subsequent manufacturing processes. Basically, the same module has been the subject of other publications, namely Grenz et al. [104] and Ostermann et al. [105]. The latter analyzed a variant containing carbon fiber-reinforced plastics, but even then, the input materials still had a greater impact on the carbon footprint than the production of the final component (via casting, forming, joining, coating, etc.). Delogu et al. [106] discussed the environmental impact of two different composites suitable for automotive dashboard panels. In all categories, with regard to GHG emissions, the raw materials phase had a higher impact than the manufacturing phase, and raw material extraction caused about three times as much emission as further processing. This was similar to car floor pans, where every kg CO_2 -eq that arose in manufacturing was accompanied by six kg CO_2 -eq during resource extraction [107]. Moreover, a comparison of different material compositions of vehicle's body and chassis by Raugei et al. [108] showed that raw materials influenced GHG emissions much more than manufacturing.

One of the most widely analyzed components in scientific publications are batteries, especially those used in electric vehicles (an overview of previous LCAs on this subject can be found in Dieterle et al. [109]; Erakca et al. [110]; and Li et al. [111]). However, few studies make a clear distinction between the emission sources along the manufacturing chain. Ellingsen et al. [112] outlined the variations found within the published studies of the time and reported on different contributions from raw material acquisition, manufacturing, and assembly. Only in 4 out of the 14 detailed carbon footprints, cell manufacture and pack assembly together yielded GHG emissions equal to or greater than material extraction. More recent studies such as Dai et al. [113] and Kelly et al. [114] concluded that the collective upstream production of battery materials used much more process energy, respectively emitting more GHGs than the cell production and assembly process.

This was contrasted by a study by Li et al. [115], who proved that the GHG emissions caused by the production of raw materials for all components of a diesel engine were only one to two percent higher than the emissions from manufacturing. The only case where this ratio was reversed was provided by Lee [61]. He examined a bumper where the raw material production accounted for 18% of the total product carbon footprint, whereas the in-house manufacturing counted for 70% and distribution counted for 12%.

Further studies on vehicle components, such as gasoline engines [116], ignition coils [117], or door panels [118], however, have provided only limited information about the contribution of individual production stages to GHG emissions. Nevertheless, a noticeable number of publications related to the automotive industry used LCAs to compare different systems (e.g., [119]), designs (e.g., [118,120,121]), materials (e.g., [122–128]), production techniques and machine settings (e.g., [129–131]), and/or production sites (e.g., [111,124]). These publications, in turn, confirmed the observations of Böttcher and Müller [58] regarding low-carbon operations.

Moreover, there is a substantial body of research dealing with the evaluation of recycled or bio-based materials for automotive applications (e.g., [132–137]). Similarly,

there are LCA studies on whether additive manufactured automotive components offer opportunities to reduce environmental impacts (e.g., [138,139]).

The authors are not aware of any openly accessible LCAs of vehicles or automotive components manufactured in Austria. Existing studies mainly compare different propulsion technologies in which the Austrian electric energy mix is considered for the use phase of electric vehicles (e.g., [22,140]). Additionally, Diendorfer et al. [20] and Meyer et al. [141] did not quantify GHG emissions on a product level for the Austrian automotive (supply) industry. The LCA for a side impact beam, presented by Mair-Bauernfeind et al. [142], is a prospective yet hypothetical scenario in which Austria is considered as a potential production site; however, this has no relevance due to a lack of real data. LCA-related results for Austria listed by Ladenika et al. [143] are also not directly related to the automotive industry.

In summary, most, but not all, studies to date suggest that, also at the component level, upstream activities contribute more to PCFs than the processing at the supplier's site. However, automotive-related manufacturing in Austria has not been studied in detail until now. Therefore, we analyzed three different car components finished in Austria to incorporate country-specific conditions to close this research gap.

4. Case Studies of Three Automotive Components

4.1. Common Methodology and Set-Up

The main objective of our analyses was to assess the carbon footprint associated with the production of the respective component. ISO 14067 [144] was chosen as the cornerstone standard for how to quantify the GHG emissions; thus, all PCF studies were subdivided into four interrelated phases: goal and scope definition, inventory analysis (LCI), impact assessment, and interpretation. Overall, our studies drew on twelve months of close industry cooperation with two companies from the automotive supply industry, incorporating data provided by the companies (e.g., measured energy and material consumption on machinery level, production volumes, and sales) and generated by the researchers (e.g., calculations related to transport and waste) through several shop floor visits, interviews with company contact persons and experts, and correspondence with external stakeholders along the supply chain (e.g., forwarding agents).

For all cases, the functional unit was defined as one salable unit of the considered component fabricated in Austria. All analyses were carried out cradle-to-gate, i.e., from the exploration of the raw materials to the production of the individual component for the OEM. Those steps carried out in between by the sub-suppliers and forwarding agents, as well as the treatment of waste generated during manufacturing, were also considered. As requested by VDA [85], general efforts such as the lighting and heating of the halls were also incorporated. Downstream transportation and distribution were ignored; in Cases 1 and 2, the OEM was responsible for transporting finished components. The product from Case 3 was supplied to different companies (first-tier suppliers and OEMs) so that no general destination could be determined. Also, the use stage, including use profiles and end-of-life treatment of the product, was set as irrelevant. Resource consumption needed for product development ('Engineering'), capital goods, business travel, and employee commuting was excluded because no purposive recording and allocation method was possible. Where unavoidable, allocation methods were used that either reflected an underlying physical relationship or were grounded on the proportion of the economic value (usually a share of the revenue) of the products. In accordance with ISO 14067 [144], carbon offsetting, which was also especially evident in the purchased electricity mix, was excluded. Figure 2 shows the generic product system for the cases (there were slight differences in individual cases since only Case 3 had an assembly). Elements shown in grey could not be assessed and were therefore outside the system boundary.





The product life cycle was excluded, as the focus was on production-related impacts. The system boundary describes the biosphere–technosphere relationship, where the elementary flows (primary energy and natural resources) represent necessary inputs to the technical system. The technosphere contains processes that ultimately generate emissions that migrate into the biosphere.

In the LCI phase, data were collected and validated. For this purpose, production data of the focal supplier was collected by measuring all kinds of energy and material consumption. The considered time period was the twelve months before the start of each study and was chosen to smooth out fluctuations in energy consumption (caused by the heating season). Sub-suppliers were asked to, at a minimum, fill out the input/output table ('Data Collection Format Spreadsheet') provided by VDA [85]. Transports were calculated based on information provided by forwarding agents. Resource consumption for defective products and parts needed for tests (both to go unsold) was tracked and attributed to the PCF of saleable products.

Using the product specifications, all raw materials could be traced back to their point of extraction, mainly by using data from the GaBi v2022.1 database [145]. The regionally specific datasets contain associated inputs from nature and emissions, including estimations on losses. In that respect, cut-offs could be avoided as far as possible. Materials not found in either the GaBi database or the other available databases (esp. Ecoinvent v3.8) were modeled from information available in the databases, or they were replaced by similar materials.

The subsequent impact assessment of the PCF study was used to calculate the potential climate change impact by multiplying the inventoried values by their substance- or resource-specific conversion factor. In this way, all GHG emissions inventoried in the LCI were weighted in terms of their impact intensity relative to carbon dioxide (expressed as kg CO_2 -eq). With regard to the time horizon, ISO 14067 [144] follows the advice of IPCC, which is why all the following figures refer to 100-year global warming potential.

We used the concluding interpretation to screen materials and processes to identify significant sources of GHG emissions and perform a sensitivity analysis of relevant inputs. This led to recommendations for action, mainly from an ecological perspective.

4.2. Case Study 1: Crossmember

A crossmember is a structural component connected to the vehicle's body at different points that ensures its stability and stiffness (torsion resistance). The crossmember assessed in this case is comparable to the innovative variant (for which an LCA is also available) mentioned in Maltese et al. [121], but it is simpler in design and easier to manufacture.

The crossmember was manufactured in a single production step through the use of a transfer press (Figure 3). This press was also used to manufacture other parts, which was appropriately reflected by allocation. The component was made entirely out of aluminum sheet and there was an agreement between the OEM and the supplier to use aluminum with at least ten percent recycled content. Furthermore, there were conservative specifications from the OEM on how to calculate a PCF. In addition to the 'polluter pay principle', reselling scrap metal was counted as waste disposal, even if it was recycled.





Around 30% of the aluminum used came from press remnants and was sold to a recycler. Lubricants directly applied onto the coil material were recorded, along with the (proportional) circulation lubrication of the press.

Besides the directly attributable consumption of the machine, there was also overhead consumption that arose from the infrastructure. Energy consumption for HVAC was determined based on the floor space and the throughput time required for production. Energy consumption for lighting was determined in a similar way (via the number of lamps in the related production area and the time needed to produce one component). Revenue was chosen on an apportionment basis for other aspects such as ancillary areas (e.g., offices, the canteen, and sanitary facilities) or the energy demand of jointly used forklifts and conveyors.

Data on the transport of coils and scrap metal (esp. by means of transportation, distances, frequency, and utilization), as well as internal and external warehousing (proportional energy consumption determined by the required space and length of the storage period), were also gathered.

As illustrated in Figure 4a, 97% of GHG emissions were linked to the raw material used and only 1% of emissions were generated in the production of the crossmember. The operation of the machine (including cooling) had the highest energy consumption, followed by heating. Since the aluminum coils were sourced from Scandinavia, transport contributed more to the PCF than the subsequent processing.



Figure 4. (a) Contribution of processes and stages to the PCF of the crossmember; (b) predicted GHG emission reduction potential for crossmember production.

The current PCF could be reduced by over 80% (Figure 4b). Aluminum is not only the largest contributor to PCF but also offers the greatest reduction potential. Low-carbon aluminum is made of recycled aluminum and produced with renewable energy sources, so it carries a footprint of less than 4 kg CO₂-eq or less per kilo of aluminum [146]. The changeover to emission-free energy has a minimal influence on the overall PCF but would eliminate almost all GHG emissions from the supplier's production. The major contribution would come from switching from gas to district heating, since the electricity purchased is already comparatively low on emissions due to the Austrian energy mix (which has a high share of hydropower) [147]. In the present case, electricity with an even higher proportion of renewable energy was intentionally sourced and was further enhanced by the installation of photovoltaic systems. The company's electricity consumption for the production of one crossmember amounted to about 90 g CO₂-eq in the examined time span. If the EU's energy mix were applied, the figure would be almost 169 g CO₂-eq (German energy mix: 256 g CO₂-eq; Polish energy mix: 591 g CO₂-eq [148]).

4.3. Case Study 2: Subframe

The subframe (rear-axle beam) is similar to the components analyzed by Cecchel et al. [100] and Ghosh et al. [101]. The component is composed of several parts (die-cast side parts, extruded aluminum profiles, and hydro bushings), which were manufactured by three European sub-suppliers. The final subframe consists of over 90% aluminum (alloys), with the remainder comprising steel, thermoplastics, natural rubber, and damper fluid.

In the finishing operation under consideration (Figure 5), the side parts were first welded to the profiles. This was followed by a machining process, in which approximately ten percent of the material was removed. In the final step, the bearings were pressed in. It was notable (although outside of the defined scope) that, by using pendular transportation racks (provided by OEM), the supplier did not need any packaging material for the finished subframes.

The resource consumption of the focal first-tier supplier was determined through explicit measurements and use of the manufacturing execution system. Welding consumables are especially worthy of mentioning. Intrafirm allocations followed a similar approach to that in Case Study 1. It was advantageous that the involved production line only manufactured the considered subframe. Two of the three sub-suppliers provided detailed data in the form of a PCF study or filled data collection format spreadsheets (including process flow charts). The third sub-supplier's components were evaluated on the basis of documents such as technical drawings and secondary data.





Figure 6a shows a breakdown of the subframe's PCF. Since the side panels are the heaviest part of the subframe and were manufactured in a resource-intensive process, they accounted for 81% of total GHG emissions. The bushings accounted for ten percent of the PCF, while the profiles had a share of seven percent. The GHG emissions of the first-tier supplier were broken down as follows: milling 71%, welding and press-in operations individually 6%, and infrastructure 17%. For the sake of completeness, it is worth mentioning that a second study was carried out according to the OEM's interpretation guidelines. In this study, the contribution for raw material turned out to be even higher.





Just as in Case Study 1, aluminum was the biggest driver of emissions. By switching to CO_2 -reduced aluminum, the PCF could be reduced by 40% or more. At the first-tier supplier, purchased electricity was the biggest lever. District heating and zero-emission electricity could reduce emissions by 86% (approximately 1% of PCF). Interestingly, if the EU's energy mix was used instead of the actual electricity mix, then 7.2 kg CO_2 -eq would be generated instead of 4.1 kg CO_2 -eq (German energy mix: 11.7 kg CO_2 -eq; Polish energy mix: 527.1 kg CO_2 -eq [148]).

The scrap metal ratio of machining was higher than that reported in Cecchel et al. [100]. However, energy consumption was lower, even if the machining cycle time was longer. Therefore, the machining time (and thus power consumption) and the amount of scrap metal certainly offered further saving potentials (a fact that was not further investigated). At least the use of minimum quantity lubrication had a positive effect on the PCF. Also left out were the optimization potentials of the product design, which could not be implemented in an economically profitable manner afterwards.

4.4. Case Study 3: Actuator

The considered actuator is a small drive located at the front bumper of a vehicle which performs the opening and closing movement of flaps to control the airflow to cool the engine and/or adjust the air resistance. It was produced in large quantities and delivered directly to both OEMs and first-tier suppliers. One actuator consists of 20 individual parts, 8 of which were manufactured in-house by injection molding. The complete product was made primarily of a thermoplastic polymer that was partially glass-fiber reinforced. Due to their small size and lightweight material, most components' masses were hardly measurable.

In-house manufacturing just included injection molding (but could be extended to also produce printed circuit boards (PCBs)). Purchased parts and in-house production parts were assembled through full automation. Finished products were first checked and then packed on trays before being palletized for shipment. The processes and parts are depicted in Figure 7.



Figure 7. Flow chart for the actuator production.

Again, the same allocation procedures were applied as in all other cases. In this case, however, it was possible to use the number of storage spaces occupied by the product and related material instead of using a share of the storage area to allocate energy consumption for warehousing. The state-of-the-art heating system was climate neutral (i.e., no GHG emissions were detectable) and was not considered in the impact assessment.

Figure 8a portrays the distribution of GHG emissions. During the period under consideration, 55% of the total PCF was embodied in granulate material and 31% was traceable to purchased parts (for which the material input was also a major influencing factor). Because of the larger number of sub-suppliers and the longer distances (some parts were manufactured in Asia), the share of transport was found to be high in comparison to Cases 1 and 2. In-house production accounted for nine percent of the total PCF. More than half of the internal GHG emissions were caused by waste (mainly sprues and rejects), that is to say, by material once again. Approximately 20% of each was attributable to the energy consumption of actual production (manufacturing and assembly), as well as infrastructure. Once again, with regard to 17.5 g CO₂-eq, a comparatively low amount of GHG was emitted (EU27: 33.3 g CO₂-eq; Germany: 48.7 g CO₂-eq; Poland: 100.9 g CO₂-eq [148]).



Figure 8. (a) Contribution of processes and stages to the PCF of the actuator; (b) predicted GHG emission reduction potential for actuator production.

At least for some parts, the glass fiber-reinforced polyamides can be replaced by other plastics. Suitable materials include carbon fiber-reinforced polypropylene and bio-based materials (as demonstrated, for example, by [134]). Rough estimates indicate a savings potential of 20%. The planned replacement of the injection molding machines to ones that feature direct injection and vacuum pumps will significantly increase efficiency. Considerably less energy (both electricity and compressed air) is consumed while, simultaneously, material is saved by avoiding sprues. Shifting PCBs from external to internal production will save energy and reduce the amount of packaging waste. Finally, an intended photovoltaic system can cover one third of the site's electricity demands. While this only slightly reduces the total PCF, it nonetheless results in a significant reduction to the corporate carbon footprint.

4.5. Comparison of Cases

All three of the previously described case studies have the following in common: they examine direct and large-scale production (the annual output volume of several tens of thousands of units) in Austria and verify that the upstream activities (i.e., raw and precursor materials) are the largest contributor of an Austrian automotive supplier's PCF. There are two reasons why the considered in-house production contributed only a single-digit percentage to the PCF. First, the analyzed factories are up to modern (energy) standards. Second, Austria has an energy mix with a very low GHG emission intensity [148]. Therefore, in two of the three case studies, building infrastructure (HVAC and lighting) accounts for only a small share of GHG emissions. Dimensions, as reported by Gebler et al. [97] and Sullivan et al. [52], could not be observed (at least not when broken down to the product level).

Table 3 compares the three components based on product complexity, proportion of in-house production, the key drivers of the PCF, and the internal PCF reduction potential, as well as the PCF reduction potential of sourcing at the overall product level. Except for the key drivers, an ordinal scale was used for categorization.

Component	Complexity	Level of In-House Production	Key Drivers	Internal Potential	Potential via Sourcing
crossmember subframe	very low low	very low low	material material material,	low low	very high high
actuator	medium	medium	production technology	medium	medium

Table 3. Cross-case analysis.

The crossmember is a very simple product that requires only one production step in-house. However, the press involved is also used for the production of other components, which complicates the measurement, calculation, and allocation of input and output. Although the subframe consists of several parts and requires more processing steps, which increases the product complexity, the data collection and allocation is simple, since the production line only manufactures this component. Due to the high number of parts and sub-suppliers involved, the actuator has a higher complexity. In addition, there is a considerable proportion of in-house production, which makes the machinery used to another important driver and lever of the PCF.

With regard to optimization potential, two factors are always considered: an efficient use of sustainably produced materials and the switch to emission-free energy. Both aspects have already been covered in the literature (e.g., [104]). From a production perspective, changing to a secondary or even bio-based material is easy and fast to implement since (usually) no adjustments to the machines are necessary. However, the experience from these three cases showed that the high requirements placed on the materials (e.g., weldability, machinability, and durability, as well as strength and elasticity) and the concerns and formalities of the OEMs complicated the change of materials. No less crucial is the OEM's willingness to pay for low-carbon components. Although economic aspects such as surcharges were intentionally ignored in the case studies, interviewed stakeholders mentioned that the vehicle sector is so competitive and its margins so low [149] that OEMs would generally not accept extra costs. Moreover, sustainable manufacturing as a unique selling proposition may also not allow for the passing on of additional costs to the customer [150]. On the other hand, some research [151–153] predict that there will be little additional cost if a vehicle is made from low-carbon materials, which makes our recommendations for a transition in sourcing more feasible. However, a basic prerequisite is that the demand for this material can be met (e.g., through sufficient recycling rates (see the prognosis of the International Aluminium Institute) [154]).

4.6. Discussion and Implications of Findings

All case studies show that industrial manufacturing in Austria emits comparatively little GHG and contributes only a small part to the PCF. The results suggest that, for simple components with little in-house production, sourcing has particularly large potential. As the proportion of in-house production increases, internal PCF reduction potentials become more important. Our findings resonate with the literature regarding environmentally responsible procurement [59,60]. They underline the importance of sourcing low-emission materials and its GHG-saving potential at the product level. However, in contrast to the majority of the publications mentioned in Section 3, the PCF reduction potential identified in the cases did not result from fundamental different materials or manufacturing processes but from quickly implementable adjustments.

The presented approach offers basic guidance when manufacturers want to identify and evaluate their product's impact and associated reduction potentials. Management can not only learn from this for upcoming requirements by legislation and OEMs but can also become aware of drivers and (further) reduction potentials of PCFs. Our study also provides implications for academia, as it addresses the research gap regarding PCF related to Austrian industry.

The predicted GHG emission reduction potentials have at least two limitations. Firstly, in only one of the three studies we considered the best available technology, and, even then, no exhaustive technology screening was conducted. Associated probability analyses (especially calculations on payback periods) were not made. Furthermore, measures to increase efficiency, e.g., through optimized machine settings, were not assessed. Secondly, our studies were by nature limited to a specific time period. A sensitivity analysis that shows how PCFs and potentials depend on production volume and capacity utilization is still to be conducted.

17 of 24

A further, yet difficult to quantify, finding was the observed 'polluter pays principle'. It could discourage suppliers from investing in low-carbon designs, processes, and materials since they will not receive any benefits for being proactive.

5. Conclusions and Outlook

One third of GHG emissions are attributable to industry and, contrary to other sectors, have hardly improved in recent years. The climate targets of developed countries such as Austria are important. However, it must not be forgotten that production (especially raw material extraction and the manufacturing of parts) has been outsourced to countries that produce more at lower costs but not necessarily at lower GHG emissions. Besides upcoming EU regulations, we brought up the net-zero pledges of automobile manufacturers. For a long time, the automotive industry has largely focused on reducing tailpipe emissions, but with breakthroughs in electric vehicles, the balance of the sector's carbon footprint is shifting to ecological vehicle production.

Since a vehicle uses resources and generates emissions before ever having driven its first kilometer, LCAs are used to determine the total harmful impact on the environment. By doing so, the true extent of the environmental load caused by the automotive industry becomes clear: it consumes one fifth of the world's aluminum and steel production and large quantities of other materials. This paper demonstrates that the majority of a vehicle production's 'carbon rucksack' can be traced back to suppliers and even further back to activities close to the 'cradle'. The success of OEMs' responses to climate change is therefore heavily dependent on the performance of their supply chain and the sourcing within.

Although numerous LCAs and PCF studies of vehicle components have been published in recent years, there is a lack of comparable studies with production sites in Austria. This paper is the first step in expanding the existing literature and addressing the research gap. Our three cases could demonstrate that the embodied emissions of (imported) materials affect the PCF much more than the finishing carried out by the supplier companies located in Austria. In all case studies, the final production stage(s) accounted for less than ten percent of the total PCF. In brief, we conclude the following: first, the substitution of current materials with low-carbon materials (i.e., recycled, bio-based, or renewable energy produced materials) is the most effective lever for reducing PCFs and assigns great responsibility to procurement. From the present project experience, we also conclude that the responsibility for ecologically sustainable production has been passed from OEMs to the suppliers. Second, within a factory, electricity is of particular importance. Even in a country such as Austria, a switch to emission-free energy sources will accomplish a significant effect. Thirdly, to be able to conduct PCF studies, there is still a need for qualification.

As a practical contribution, our findings can be used by companies to prepare for future PCF reporting. The illustrated reduction potentials have been perceived as useful (in achieving potential market advantages) by representatives of the involved companies, especially if PCFs are to become a sourcing decision criterion. In analogy to Lee (2011) and to ensure (future) applicability of our approach, step-by step guidelines and individual templates for recording resource consumption have been developed for all cases, with a particular focus on assisting allocation procedures. In addition, the experience gained will not only support the companies involved but will also be used in the engineering education at Graz University of Technology. We believe it is of particular importance to raise awareness of the relevance of upstream activities as well as the challenges in defining system boundaries and allocating resource consumptions.

Based on our findings, future research should consider additional cases and focus on aspects outside the analyzed scope, such as the excluded engineering activities, best available technologies, or economic viability of decarbonization measures. Additionally, further research is required to address technical limitations such as the limited availability of sustainably produced materials. **Author Contributions:** Conceptualization, M.W.; Investigation, K.R. and M.W.; Validation, K.R.; Writing—original draft, K.R.; Writing—review & editing, M.W.; Visualization, K.R. and M.W.; Supervision, M.W. All authors have read and agreed to the published version of the manuscript.

Funding: This publication was supported by the Open Access Funding by the Graz University of Technology.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: The data used to support the findings of this study are available from the corresponding author upon request.

Acknowledgments: The authors would like to thank the representatives of both companies for authorizing the use of data for scientific purposes. We would also like to thank Sila Temizel-Sekeryan and Manuel Ulrich for their invaluable support.

Conflicts of Interest: The authors declare that they have no known competing financial interest or personal relationship that could have appeared to influence this article. We also did not receive any specific grant from funding agencies in the public, commercial, or not-for-profit sectors.

Abbreviations

- ETS Emissions Trading System
- EU European Union
- GHG greenhouse gas
- GRI Global Reporting Initiative
- GVA gross value added
- HVAC heating, ventilation, air conditioning
- IPCC Intergovernmental Panel on Climate Change
- ISIC International Standard Industrial Classification (of all economic activities)
- LCA life cycle assessment
- LCI life cycle inventory (analysis)
- OECD Organization for Economic Co-operation and Development
- OEM original equipment manufacturer
- PCF product carbon footprint
- VDA German Association of the Automotive Industry (Verband der Automobilindustrie)

References

- 1. International Monetary Fund. Climate Change Indicators Dashboard. 2023. Available online: https://climatedata.imf.org/pages/ access-data (accessed on 19 July 2023).
- Ritchie, H.; Roser, M.; Rosado, P. CO₂ and Greenhouse Gas Emissions. 2020. Available online: https://ourworldindata.org/co2
 -and-greenhouse-gas-emissions (accessed on 21 August 2023).
- Lynas, M.; Houlton, B.; Simon, P. Greater than 99% consensus on human caused climate change in the peer-reviewed scientific literature. *Environ. Res. Lett.* 2021, 16, 114005. [CrossRef]
- 4. Myers, K.; Doran, P.; Cook, J.; Kotcher, J.; Myers, T. Consensus revisited: Quantifying scientific agreement on climate change and climate expertise among Earth scientists 10 years later. *Environ. Res. Lett.* **2021**, *16*, 104030. [CrossRef]
- 5. OECD.Stat. 2023. Available online: https://stats.oecd.org/ (accessed on 21 August 2023).
- Arto, I.; Dietzenbacher, E. Drivers of the growth in global greenhouse gas emissions. *Environ. Sci. Technol.* 2014, 48, 5388–5394. [CrossRef] [PubMed]
- 7. Peters, G. Carbon footprints and embodied carbon at multiple scales. *Curr. Opin. Environ. Sustain.* 2010, 2, 245–250. [CrossRef]
- European Commission. Proposal for a Regulation of the European Parliament and of the Council Establishing a Carbon Border Adjustment Mechanism. 2021. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/?uri=celex:52021PC0564 (accessed on 21 August 2023).
- 9. Fang, Y.; Yu, Y.; Shi, Y.; Liu, J. The effect of carbon tariffs on global emission control: A global supply chain model. *Transp. Res. Part E Logist. Transp. Rev.* **2020**, *133*, 101818. [CrossRef]
- Bashmakov, I.; Nilsson, L.; Acquaye, A.; Bataille, C.; Cullen, J.; de la Rue du Can, S.; Fischedick, M.; Geng, Y.; Tanaka, K. Industry. In *Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*; Cambridge University Press: Cambridge, UK, 2022. [CrossRef]

- 11. European Council. Fit for 55. 2023. Available online: https://www.consilium.europa.eu/en/policies/green-deal/fit-for-55-theeu-plan-for-a-green-transition/ (accessed on 21 August 2023).
- European Commission. The Green Deal Industrial Plan. 2023. Available online: https://commission.europa.eu/strategy-and-policy/priorities-2019-2024/european-green-deal/green-deal-industrial-plan_en (accessed on 21 August 2023).
- 13. European Commission. Report from the Commission to the European Parliament and the Council on the Functioning of the European Carbon Market in 2021 Pursuant to Articles 10(5) and 21(2) of Directive 2003/87/EC. 2022. Available online: https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:52022DC0516 (accessed on 21 August 2023).
- 14. Hale, T.; Smith, S.; Black, R.; Cullen, K.; Fay, B.; Lang, J.; Mahmood, S. Assessing the rapidly-emerging landscape of net zero targets. *Clim. Policy* **2022**, *22*, 18–29. [CrossRef]
- 15. NetZeroTracker. Data Explorer. 2023. Available online: https://zerotracker.net/ (accessed on 21 August 2023).
- 16. Republic of Austria. Aus Verantwortung für Österreich. Regierungsprogramm 2020–2024. 2020. Available online: https://www.dievolkspartei.at/Download/Regierungsprogramm_2020.pdf (accessed on 21 August 2023). (In German)
- Kirchengast, G.; Steininger, K.; Schleicher, S. Klimaschutzzielpfad f
 ür Österreich. 2021. Available online: https://wegccloud.unigraz.at/s/oT2QKiJgFkLdjwc (accessed on 21 August 2023). (In German)
- Federal Ministry for Climate Action, Environment, Energy, Mobility, Innovation and Technology. Fortschrittsbericht 2022 nach § 6 Klimaschutzgesetz. 2022. Available online: https://www.bmk.gv.at/dam/jcr:5589d089-dcc6-476b-bf21-d0231ceccc28/KSG-Fortschrittsbericht_2022_Layout_4.pdf (accessed on 21 August 2023). (In German)
- European Parliamentary Research Service. Climate Action in Austria. 2021. Available online: https://www.europarl.europa.eu/ RegData/etudes/BRIE/2021/696186/EPRS_BRI(2021)696186_EN.pdf (accessed on 21 August 2023).
- Diendorfer, C.; Gahleitner, B.; Dachs, B.; Kienberger, T.; Nagovnak, P.; Böhm, H.; Moser, S.; Thenius, G.; Knaus, K. Climate Neutrality for Austria by 2040—Contribution of Austrian Industry. 2021. Available online: www.bmk.gv.at/dam/jcr:77f55fa2-d9 a6-42d9-8cd0-64e466d2c913/Climate_neutrality_Austria_2040.pdf (accessed on 21 August 2023).
- 21. Hetterich, J.; Bonnemeier, S.; Pritzke, M.; Georgiadis, A. Ecological sustainability—A customer requirement? Evidence from the automotive industry. *J. Environ. Plan. Manag.* 2012, *55*, 1111–1133. [CrossRef]
- Environment Agency Austria. Die Ökobilanz von Personenkraftwagen. Report 0763. 2021. Available online: https://www. umweltbundesamt.at/fileadmin/site/publikationen/rep0763.pdf (accessed on 21 August 2023). (In German)
- 23. Azari, R.; Badri, N. Life cycle assessment as a research methodology for estimating the environmental impacts of buildings. In *Research Methods in Building Science and Technology;* Springer: Cham, Switzerland, 2021. [CrossRef]
- 24. Shaukat, M. LCA in Saudi Arabia: A critical review. Int. J. Life Cycle Assess. 2023, 28, 1419–1427. [CrossRef]
- 25. Bueno, C.; Fabricio, M. Comparative analysis between a complete LCA study and results from a BIM-LCA plug-in. *Autom. Constr.* **2018**, *90*, 188–200. [CrossRef]
- Gilardino, A.; Rojas, J.; Mattos, H.; Larrea-Gallegos, G.; Vázquez-Rowe, I. Combining operational research and life cycle assessment to optimize municipal solid waste collection in a district in Lima (Peru). J. Clean. Prod. 2017, 156, 589–603. [CrossRef]
- 27. Pandey, D.; Agrawal, M.; Pandey, J. Carbon footprint: Current methods of estimation. *Environ. Monit. Assess.* 2011, 178, 135–160. [CrossRef]
- 28. European Commission. Questions about the Product Environmental Footprint and Organisation Environmental Footprint Methods. 2023. Available online: http://ec.europa.eu/environment/eussd/smgp/pdf/q_a.pdf (accessed on 21 August 2023).
- Organisation Internationale des Constructeurs Automobiles. Production Statistics. 2023. Available online: https://www.oica.net/production-statistics/ (accessed on 21 August 2023).
- IBISWorld. Global Car & Automobile Manufacturing. 2022. Available online: https://www.ibisworld.com/global/market-size/ global-car-automobile-manufacturing/ (accessed on 24 July 2023).
- McKinsey. Outlook on the Automotive Software and Electronics Market through 2030. 2019. Available online: https://www.mckinsey.com/industries/automotive-and-assembly/our-insights/mapping-the-automotive-software-and-electronics-landscape-through-2030 (accessed on 21 August 2023).
- 32. Association des Constructeurs Européens d'Automobiles. Economic and Market Report: State of the EU Auto Industy. 2023. Available online: https://www.acea.auto/files/Economic-and-Market-Report_Full-year-2022.pdf (accessed on 1 October 2023).
- Destatis. Automotive Izndustry Largest Share of Gross Value Added. 2019. Available online: https://www.destatis.de/Europa/ EN/Topic/Industry-trade-services/Industry.html (accessed on 21 August 2023).
- Slacik, T.; Walko, Z. Automotive Industry and Global Value Chain Integration in CESEE in COVID-Times and beyond. 2020. Available online: https://www.oenb.at/dam/jcr:fdad9d36-8944-4440-9845-2dcce93fd57d/09_Konjunktur-aktuell_12_20_Automotive-industry.pdf (accessed on 21 August 2023).
- Statistik Austria. Structural Business Statistics. 2023. Available online: https://www.statistik.at/en/statistics/industryconstruction-trade-and-services/structural-business-statistics (accessed on 21 August 2023).
- Statistik Austria. Gross Domestic Product and Main Aggregates. 2023. Available online: https://www.statistik.at/en/statistics/ national-economy-and-public-finance/national-accounts/gross-domestic-product-and-main-aggregates (accessed on 21 August 2023).
- Fachverband der Fahrzeugindustrie Österreichs. Die Österreichische Fahrzeugindustrie. 2022. Available online: https://www. fahrzeugindustrie.at/fileadmin/content/Zahlen___Fakten/Wirtschaftsfaktor_Automobil/Seite1.3_-_1.8_2022.pdf (accessed on 21 August 2023). (In German)

- Statistik Austria. NAMEA. 2023. Available online: https://www.statistik.at/en/statistics/energy-and-environment/ environment/namea (accessed on 21 August 2023).
- Jaramillo, P.; Kahn Ribeiro, S.; Newman, P.; Dhar, S.; Diemuodeke, O.; Kajino, T.; Lee, D.; Nugroho, S.; Ou, X.; Hammer Strømman, A.; et al. Transport. In *Climate Change 2022: Mitigation of Climate Change. Contribution of Working Group III to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*; Cambridge University Press: Cambridge, UK, 2022. [CrossRef]
- 40. Sprei, F. Disrupting mobility. Energy Res. Soc. Sci. 2018, 37, 238-242. [CrossRef]
- 41. 42 USC § 17111. Future of Industry Program. Available online: https://www.law.cornell.edu/uscode/text/42/17111 (accessed on 21 August 2023).
- 42. Energy Information Administration. Industrial sector energy consumption. In *International Energy Outlook 2016;* Energy Information Administration: Washington, DC, USA, 2016. Available online: https://www.eia.gov/outlooks/ieo/pdf/industrial.pdf (accessed on 21 August 2023).
- 43. European Commission. *Masterplan for a Competitive Transformation of EU Energy-Intensive Industries Enabling a Climate-Neutral, Circular Economy by* 2050; Publications Office of the EU: Luxembourg, Germany, 2019. [CrossRef]
- EUROFER. European Steel in Figures 2023. 2023. Available online: https://www.eurofer.eu/assets/publications/brochuresbooklets-and-factsheets/european-steel-in-figures-2023/FINAL_EUROFER_Steel-in-Figures_2023.pdf (accessed on 21 August 2023).
- 45. Worldsteel. Steel in Automotive. 2023. Available online: https://worldsteel.org/steel-topics/steel-markets/automotive/ (accessed on 21 August 2023).
- Wormington, J.; Bugalski, N.; Scott, C. Aluminum: The Car Industry's Blind Spot. Human Rights Watch & Inclusive Development International. 2021. Available online: https://www.hrw.org/sites/default/files/media_2021/10/global_bauxite0721_web.pdf (accessed on 21 August 2023).
- 47. European Aluminium. Aluminum Content in Passenger Vehicles (Europe). 2023. Available online: https://european-aluminium. eu/wp-content/uploads/2023/05/23-05-02Aluminum-Content-in-Cars_Public-Summary.pdf (accessed on 21 August 2023).
- Glass Alliance Europe. Panorama of the EU Glass Industry. 2023. Available online: https://www.glassallianceeurope.eu/images/ cont/panorama-2022-eu27_1_file.pdf (accessed on 24 July 2023).
- Glass for Europe. Key Data. 2023. Available online: https://glassforeurope.com/the-sector/key-data/ (accessed on 21 August 2023).
- 50. Sato, F.; Nakata, T. Energy consumption analysis for vehicle production through a material flow approach. *Energies* **2020**, *13*, 2396. [CrossRef]
- 51. Li, Y.; Fujikawa, K.; Wang, J.; Li, X.; Ju, Y.; Chen, C. The potential and trend of end-of-life passenger vehicles recycling in China. *Sustainability* **2020**, *12*, 1455. [CrossRef]
- 52. Sullivan, J.; Burnham, A.; Wang, M. Energy-Consumption and Carbon-Emission Analysis of Vehicle and Component Manufacturing; Argonne National Laboratory: Argonne, IL, USA, 2010. [CrossRef]
- 53. World Economic Forum. Forging Ahead: A Materials Roadmap for the Zero-Carbon Car. 2020. Available online: https://www3.weforum.org/docs/WEF_Forging_Ahead_2020.pdf (accessed on 21 August 2023).
- 54. Hammond, G.; Jones, C. Embodied Carbon—The Inventory of Carbon and Energy (ICE); BSIRA: Bracknell, UK, 2011.
- 55. Hampshire-Waugh, M. Climate Change and the Road to Net-Zero; Crowstone Publishing: London, UK, 2021.
- 56. Damert, M.; Baumgartner, R. Intra-sectoral differences in climate change strategies: Evidence from the global automotive industry. *Bus. Strategy Environ.* **2018**, 27, 265–281. [CrossRef] [PubMed]
- 57. Nolan, P.; Zhang, J.; Liu, C. The Global Business Revolution and the Cascade Effect; Palgrave Macmillan: London, UK, 2007. [CrossRef]
- 58. Böttcher, C.; Müller, M. Drivers, practices and outcomes of low-carbon operations: Approaches of German automotive suppliers to cutting carbon emissions. *Bus. Strategy Environ.* **2015**, *24*, 477–498. [CrossRef]
- Correia, F.; Howard, M.; Hawkins, B.; Pye, A.; Lamming, R. Low carbon procurement: An emerging agenda. J. Purch. Supply Manag. 2013, 19, 58–64. [CrossRef]
- 60. Sarkis, J.; Dou, Y. Green Supply Chain Management; Routledge: New York, NY, USA, 2017. [CrossRef]
- 61. Lee, K.-H. Integrating carbon footprint into supply chain management: The case of Hyundai Motor Company (HMC) in the automobile industry. *J. Clean. Prod.* 2011, *19*, 1216–1223. [CrossRef]
- 62. Azevedo, S.; Barros, M. The application of the triple bottom line approach to sustainability assessment: The case study of the UK automotive supply chain. *J. Ind. Eng. Manag.* **2017**, *10*, 286–322. [CrossRef]
- 63. Stone, K. Four decades of lean: A systematic literature review. Int. J. Lean Six Sigma 2012, 3, 112–132. [CrossRef]
- 64. Helmold, M.; Küçük Yılmaz, A.; Flouris, T.; Winner, T.; Cvetkoska, V.; Dathe, T. Lean management in the automotive industry. In *Lean Management, Kaizen, Kata and Keiretsu*; Springer: Cham, Switzerland, 2022. [CrossRef]
- 65. Cronite. Environmental Study. 2020. Available online: https://www.cronite-group.com/environmental-study_text/#Last-Wax (accessed on 21 August 2023).
- Statista Market Insights. Passenger Cars—Worldwide. 2023. Available online: https://www.statista.com/outlook/mmo/ passenger-cars/worldwide (accessed on 21 August 2023).
- 67. Environmental Protection Agency. Automotive Trends Report—Executive Summary. 2022. Available online: https://www.epa.gov/system/files/documents/2022-12/420s22001.pdf (accessed on 21 August 2023).

- Liu, W.; Hao, J.; Kong, A. Breaking the Mold: The Role of Automakers in Steel Decarbonization. 2023. Available online: https: //www.greenpeace.org/static/planet4-eastasia-stateless/2023/05/e45b70a3-auto_steel_report_2023_-greenpeace.pdf (accessed on 21 August 2023).
- 69. Bertram, M.; Buxmann, K.; Furrer, P. Analysis of greenhouse gas emissions related to aluminium transport applications. *Int. J. Life Cycle Assess.* **2009**, 14 (Suppl. S1), 62–69. [CrossRef]
- European Aluminium. Circular Aluminium Action Plan—Executive Summary. 2020. Available online: https://europeanaluminium.eu/wp-content/uploads/2022/08/2020-05-13_european-aluminium_circular-aluminium-action-plan_executivesummary.pdf (accessed on 21 August 2023).
- International Aluminium Institute. IAI Material Flow Model—2021 Update. 2021. Available online: https://internationalaluminium.org/wp-content/uploads/2021/05/IAI-Material-Flow-Model-2021-Update.pdf (accessed on 24 July 2023).
- 72. American Iron and Steel Institute. Profile. 2023. Available online: https://www.steel.org/wp-content/uploads/2023/04/AISI-Profile-Book_UPDATED-Mar-2023-REDUCED-SIZE.pdf (accessed on 21 August 2023).
- 73. Cui, J.; Roven, H. Recycling of automotive aluminum. Trans. Nonferrous Met. Soc. China 2010, 20, 2057–2063. [CrossRef]
- 74. Pietroluongo, M.; Padovano, E.; Frache, A.; Badini, C. Mechanical recycling of an end-of-life automotive composite component. *Sustain. Mater. Technol.* **2020**, *23*, e00143. [CrossRef]
- 75. Ravina, M.; Bianco, I.; Ruffino, B.; Minardi, M.; Panepinto, D.; Zanetti, M. Hard-to-recycle plastics in the automotive sector: Economic, environmental and technical analyses of possible actions. *J. Clean. Prod.* **2023**, *394*, 136227. [CrossRef]
- 76. Sukitsch, M.; Engert, S.; Baumgartner, R. The implementation of corporate sustainability in the European automotive industry: An analysis of sustainability reports. *Sustainability* **2015**, *7*, 11504–11531. [CrossRef]
- 77. Lukin, E.; Krajnović, A.; Bosna, J. Sustainability strategies and achieving SDGs: A comparative analysis of leading companies in the automotive industry. *Sustainability* **2022**, *14*, 4000. [CrossRef]
- GRI. Four-in-Five Largest Global Companies Report with GRI. 2022. Available online: https://www.globalreporting.org/news/ news-center/four-in-five-largest-global-companies-report-with-gri/ (accessed on 24 July 2023).
- GRI. Consolidated Set of the GRI Standards. 2022. Available online: https://www.globalreporting.org/how-to-use-the-gristandards/gri-standards-english-language/ (accessed on 24 July 2023).
- Caliskan, A.; Ozturkoglu, O.; Ozturkoglu, Y. Ranking of responsible automotive manufacturers according to sustainability reports using PROMETHEE and VIKOR methods. *Adv. Sustain. Syst.* 2022, *6*, 2100301. [CrossRef]
- 81. BMW. CO₂ Reduction. 2023. Available online: https://www.bmwgroup.com/en/sustainability/our-focus/co2-reduction.html (accessed on 21 August 2023).
- Toyota. Sustainability Data Book 2002. 2022. Available online: https://global.toyota/pages/global_toyota/sustainability/report/sdb/sdb22_en.pdf (accessed on 24 July 2023).
- Scipioni, A.; Manzardo, A.; Mazzi, A.; Mastrobuono, M. Monitoring the carbon footprint of products: A methodological proposal. J. Clean. Prod. 2012, 36, 94–101. [CrossRef]
- Finkbeiner, M.; Krinke, S.; Oschmann, D.; Saeglitz, T.; Schaper, S.; Schmidt, W.-P.; Schell, R. Data collection format for life cycle assessment of the German Association of the Automotive Industry (VDA). *Int. J. Life Cycle Assess.* 2003, *8*, 379–381. [CrossRef]
- VDA. Life Cycle Assessment. 2007. Available online: https://www.vda.de/en/news/publications/publication/life-cycleassessment (accessed on 21 August 2023).
- Huang, Y.; Weber, C.; Matthews, H. Categorization of scope 3 emissions for streamlined enterprise carbon footprinting. *Environ.* Sci. Technol. 2009, 43, 8509–8515. [CrossRef]
- Schmidt, M.; Nill, M.; Scholz, J. Determining the scope 3 emissions of companies. *Chem. Eng. Technol.* 2022, 45, 1218–1230. [CrossRef]
- GHG Protocol. Technical Guidance for Calculating Scope 3 Emissions; World Resources Institute & World Business Council for Sustainable Development: Geneva, Switzerland, 2013; Available online: https://ghgprotocol.org/sites/default/files/2023-03/ Scope3_Calculation_Guidance_0[1].pdf (accessed on 24 July 2023).
- 89. Forbes. The Global 2000. 2023. Available online: https://www.forbes.com/lists/global2000/ (accessed on 24 July 2023).
- BMW. BMW Group Report 2022. 2023. Available online: https://www.bmwgroup.com/content/dam/grpw/websites/ bmwgroup_com/ir/downloads/en/2023/bericht/BMW-Group-Report-2022-en.pdf (accessed on 24 July 2023).
- Ford. Integrated Sustainability and Financial Report 2023. 2023. Available online: https://corporate.ford.com/content/dam/ corporate/us/en-us/documents/reports/2023-integrated-sustainability-and-financial-report.pdf (accessed on 24 July 2023).
- GM. 2022 Sustainability Report—Data Center. 2023. Available online: https://www.gmsustainability.com/_pdf/resources-anddownloads/GM_ESG_Data_2022.pdf (accessed on 24 July 2023).
- Hyundai. 2023 Sustainability Report. 2023. Available online: https://www.hyundai.com/content/hyudai/ww/data/csr/data/ 0000000051/attach/english/hmc-2023-sustainability-report-en.pdf (accessed on 19 July 2023).
- Mercedes-Benz. Sustainability Report 2022. 2023. Available online: https://sustainabilityreport.mercedes-benz.com/2022/_ assets/downloads/entire-mercedes-benz-sr22.pdf (accessed on 24 July 2023).
- 95. Stellantis. 2022 Corporate Social Responsibility Report. 2023. Available online: https://www.stellantis.com/content/dam/ stellantis-corporate/sustainability/csr-disclosure/stellantis/2022/Stellantis-2022-CSR-Report.pdf (accessed on 24 July 2023).
- 96. VW. Group Sustainability Report 2022. 2023. Available online: https://www.volkswagen-group.com/en/publications/more/ group-sustainability-report-2022-1644 (accessed on 24 July 2023).

- 97. Gebler, M.; Cerdas, J.; Thiede, S.; Herrmann, C. Life cycle assessment of an automotive factory: Identifying challenges for the decarbonization of automotive production—A case study. *J. Clean. Prod.* **2020**, *270*, 122330. [CrossRef]
- 98. Rivera, J.; Reyes-Carrillo, T. A life cycle assessment framework for the evaluation of automobile paint shops. J. Clean. Prod. 2016, 115, 75–87. [CrossRef]
- 99. Kim, H.; Wallington, T. Life-cycle energy and greenhouse gas emission benefits of lightweighting in automobiles: Review and harmonization. *Environ. Sci. Technol.* **2013**, *47*, 6089–6097. [CrossRef] [PubMed]
- Cecchel, S.; Cornacchia, G.; Panvini, A. Cradle-to-gate impact assessment of a high-pressure die-casting safety-relevant automotive component. JOM 2016, 68, 2443–2448. [CrossRef]
- Ghosh, T.; Kim, H.; De Kleine, R.; Wallington, T.; Bakshi, B. Life cycle energy and greenhouse gas emissions implications of using carbon fiber reinforced polymers in automotive components: Front subframe case study. *Sustain. Mater. Technol.* 2021, 28, e00263. [CrossRef]
- 102. Del Pero, F.; Delogu, M.; Kerschbaum, M. Design of a lightweight rear crash management system in a sustainable perspective. *Sustainability* **2020**, *12*, 5243. [CrossRef]
- 103. Del Pero, F.; Delogu, M.; Pierini, M.; Kerschbaum, M.; Toelle, J. *Design and Sustainability Assessment of Lightweight Concept for an Automotive Car Module*; SAE Technical Paper 2020-37-0033; SAE: Warrendale, PA, USA, 2020. [CrossRef]
- Grenz, J.; Ostermann, M.; Käsewieter, K.; Cerdas, F.; Marten, T.; Herrmann, C.; Tröster, T. Integrating prospective LCA in the development of automotive components. *Sustainability* 2023, 15, 10041. [CrossRef]
- Ostermann, M.; Grenz, J.; Triebus, M.; Cerdas, F.; Marten, T.; Tröster, T.; Herrmann, C. Integrating prospective scenarios in life cycle engineering: Case study of lightweight structures. *Energies* 2023, 16, 3371. [CrossRef]
- Delogu, M.; Zanchi, L.; Maltese, S.; Bonoli, A.; Pierini, M. Environmental and economic life cycle assessment of a lightweight solution for an automotive component: A comparison between talc-filled and hollow glass microspheres-reinforced polymer composites. J. Clean. Prod. 2016, 139, 548–560. [CrossRef]
- Raugei, M.; El Fakir, O.; Wang, L.; Lin, J.; Morrey, D. Life cycle assessment of the potential environmental benefits of a novel hot forming process in automotive manufacturing. *J. Clean. Prod.* 2014, 83, 80–86. [CrossRef]
- Raugei, M.; Morrey, D.; Hutchinson, A.; Winfield, P. A coherent life cycle assessment of a range of lightweighting strategies for compact vehicles. J. Clean. Prod. 2015, 108 Pt A, 1168–1176. [CrossRef]
- Dieterle, M.; Fischer, P.; Pons, M.-N.; Blume, N.; Minke, C.; Bischi, A. Life cycle assessment (LCA) for flow batteries: A review of methodological decisions. Sustain. Energy Technol. Assess. 2022, 53 Pt A, 102457. [CrossRef]
- 110. Erakca, M.; Pinto Bautista, S.; Moghaddas, S.; Baumann, M.; Bauer, W.; Leuthner, L.; Weil, M. Closing gaps in LCA of lithium-ion batteries: LCA of lab-scale cell production with new primary data. *J. Clean. Prod.* **2023**, *384*, 135510. [CrossRef]
- 111. Li, P.; Xia, X.; Guo, J. A review of the life cycle carbon footprint of electric vehicle batteries. *Sep. Purif. Technol.* **2022**, 296, 121389. [CrossRef]
- 112. Ellingsen, L.; Hung, C.; Strømman, A. Identifying key assumptions and differences in life cycle assessment studies of lithium-ion traction batteries with focus on greenhouse gas emissions. *Transp. Res. Part D Transp. Environ.* **2017**, *55*, 82–90. [CrossRef]
- Dai, Q.; Kelly, J.; Gaines, L.; Wang, M. Life cycle analysis of lithium-ion batteries for automotive applications. *Batteries* 2019, 5, 48. [CrossRef]
- 114. Kelly, J.; Wang, M.; Dai, Q.; Winjobi, O. Energy, greenhouse gas, and water life cycle analysis of lithium carbonate and lithium hydroxide monohydrate from brine and ore resources and their use in lithium ion battery cathodes and lithium ion batteries. *Resour. Conserv. Recycl.* **2021**, 174, 105762. [CrossRef]
- 115. Li, T.; Liu, Z.-C.; Zhang, H.-C.; Jiang, Q.-H. Environmental emissions and energy consumptions assessment of a diesel engine from the life cycle perspective. *J. Clean. Prod.* **2013**, *53*, 7–12. [CrossRef]
- 116. Dhingra, R.; Das, S. Life cycle energy and environmental evaluation of downsized vs. lightweight material automotive engines. *J. Clean. Prod.* 2014, *85*, 347–358. [CrossRef]
- Chang, H.-L.; Chen, C.-M.; Sun, C.-H.; Liu, H.-D. Carbon footprint of automotive ignition coil. *IOP Conf. Ser. Mater. Sci. Eng.* 2015, 87, 012016. [CrossRef]
- 118. Muñoz, I.; Rieradevall, J.; Domènech, X.; Gazulla, C. Using LCA to assess eco-design in the automotive sector: Case study of a polyolefinic door panel. *Int. J. Life Cycle Assess.* **2006**, *11*, 323–334. [CrossRef]
- 119. Warburg, N.; Mailaender, E.; Herrmann, C.; Eyerer, P.; Reckziegel, C.; Saatkamp, T.; Stutz, M. Obtaining environmental benefits by application of electronic systems-automotive LCA case study: Starter generator vs. conventional system. In Proceedings of the IEEE International Symposium on Electronic and the Environment, San Francisco, CA, USA, 6–9 May 2002; pp. 27–31. [CrossRef]
- 120. Alonso, J.; Bigorra, J.; Giró, J. Integrating LCA and DfE in the Design of Electrical & Electronic Products for the Automotive Sector; SAE Technical Paper 2001-01-3742; SAE: Warrendale, PA, USA, 2001. [CrossRef]
- 121. Maltese, S.; Delogu, M.; Zanchi, L.; Bonoli, A. Application of design for environment principles combined with LCA methodology on automotive product process development: The case study of a crossmember. In *Sustainable Design and Manufacturing* 2017; Smart Innovation, Systems and Technologies 68; Springer: Cham, Switzerland, 2017. [CrossRef]
- 122. Akhshik, M.; Panthapulakkal, S.; Tjong, S.; Sain, M. Life cycle assessment and cost analysis of hybrid fiber-reinforced engine beauty cover in comparison with glass fiber-reinforced counterpart. *Environ. Impact Assess. Rev.* 2017, 65, 111–117. [CrossRef]

- 123. Bertram, M.; Bayliss, C.; Ragnarsson, P. Updated Automotive Lifecycle Assessment Model Launched. Light Metal Age. 2019. Available online: https://european-aluminium.eu/wp-content/uploads/2022/10/lma_jun_2019_p010_updated-automotivelifecycle-assessment-model-launched-reprint-1.pdf (accessed on 21 August 2023).
- 124. Schnöll, H.; Brunner, H.; Zottler, M.; Hirz, M.; Ramsauer, C. CO₂ reduction potential of CFRP-bodywork concepts considering production and in-use phase. In Proceedings of the International Conference Management of Technology—Step to Sustainable Production: MOTSP 2015, Brela, Croatia, 10–12 June 2015.
- 125. Sun, X.; Liu, J.; Lu, B.; Zhang, P.; Zhao, M. Life cycle assessment-based selection of a sustainable lightweight automotive engine hood design. *Int. J. Life Cycle Assess.* 2017, 22, 1373–1383. [CrossRef]
- 126. Tadele, D.; Roy, P.; Defersha, F.; Mohanty, A. A comparative life-cycle assessment of talc- and biochar-reinforced composites for lightweight automotive parts. *Clean Technol. Environ. Policy* **2020**, *22*, 639–649. [CrossRef]
- 127. Tharumarajah, A.; Koltun, P. Improving environmental performance of magnesium instrument panels. *Resour. Conserv. Recycl.* **2010**, *54*, 1189–1195. [CrossRef]
- 128. Worldautosteel. Battery Electric Vehicle Life Cycle Energy Aluminium vs. AHSS: A Case Study. 2021. Available online: https: //www.worldautosteel.org/life-cycle-thinking/case-studies/battery-electric-vehicle-life-cycle-energy-aluminium-vs-ahss/ (accessed on 21 August 2023).
- Forcellese, A.; Mancia, T.; Simoncini, M.; Gentili, S.; Marconi, M.; Vita, A.; Nardinocchi, A.; Castorani, V. Comparative life cycle assessment of carbon fiber reinforced composite components for automotive industry. In Proceedings of the 24th International Conference on Material Forming, Liège, Belgium, 14–16 April 2021. [CrossRef]
- 130. Lopes Silva, D.; de Oliveira, J.; Filleti, R.; de Oliveira, J.; da Silva, E.; Ometto, A. Life cycle assessment in automotive sector: A case study for engine valves towards cleaner production. *J. Clean. Prod.* **2018**, *184*, 286–300. [CrossRef]
- 131. Soares, L.; Firmino, A.; de Oliveira, J.; Lopes Silva, D.; Saavedra, Y.; da Silva Moris, V. Comparative LCA of automotive gear hobbing processes with flood lubrication, M.Q.L. *Int. J. Adv. Manuf. Technol.* **2020**, *119*, 1071–1090. [CrossRef]
- Dunne, R.; Desai, D.; Sadiku, R.; Jayaramudu, J. A review of natural fibres, their sustainability and automotive applications. J. Reinf. Plast. Compos. 2016, 35, 1041–1050. [CrossRef]
- 133. Kumar, N.; Das, D. Fibrous biocomposites from nettle (*Girardinia diversifolia*) and poly(lactic acid) fibers for automotive dashboard panel application. *Compos. Part B Eng.* **2017**, *130*, 54–63. [CrossRef]
- Luz, S.; Caldeira-Pires, A.; Ferrão, P. Environmental benefits of substituting talc by sugarcane bagasse fibers as reinforcement in polypropylene composites: Ecodesign and LCA as strategy for automotive components. *Resour. Conserv. Recycl.* 2010, 54, 1135–1144. [CrossRef]
- 135. Pegoretti, T.; Mathieux, F.; Evrard, D.; Brissaud, D.; Arruda, J. Use of recycled natural fibres in industrial products: A comparative LCA case study on acoustic components in the Brazilian automotive sector. *Resour. Conserv. Recycl.* 2014, *84*, 1–14. [CrossRef]
- 136. Ryntz, R.; Kozora, S. Sustainable materials in automotive. *Phys. Sci. Rev.* 2019, 4, 20190021. [CrossRef]
- 137. Wakeel, S.; Bingol, S.; Bashir, M.; Ahmad, S. Selection of sustainable material for the manufacturing of complex automotive products using a new hybrid goal programming model for best worst method-proximity indexed value method. *Proc. Inst. Mech. Eng. Part L J. Mater. Des. Appl.* **2020**, 235, 385–399. [CrossRef]
- 138. Agrawal, R.; Vinodh, S. Life cycle assessment of an additive manufactured automotive component. In *Advances in Additive Manufacturing and Joining. Lecture Notes on Multidisciplinary Industrial Engineering*; Springer: Singapore, 2020. [CrossRef]
- Jaganaa, A.; Taruna, V.; Reshawantha, K.; Rajyalakshmi, G.; Jayakrishna, K. Prediction and optimization of sustainable production processes for automotive components. In *Progress in Sustainable Manufacturing*. *Management and Industrial Engineering*; Springer: Singapore, 2023. [CrossRef]
- 140. Hirz, M.; Brunner, H. Eco-design in the automotive industry—Potentials and challenges. In Proceedings of the International Conference Management of Technology—Step to Sustainable Production: MOTSP 2015, Brela, Croatia, 10–12 June 2015.
- 141. Meyer, B.; Friesenbichler, K.; Hirz, M. Dekarbonisierung als ein Treiber des Wandels der österreichischen Kfz-Zulieferindustrie. WIFO Monatsberichte 2021, 94, 829–839. Available online: https://www.wifo.ac.at/jart/prj3/wifo/resources/person_dokument/ person_dokument.jart?publikationsid=69190&mime_type=application/pdf (accessed on 21 August 2023). (In German)
- Mair-Bauernfeind, C.; Zimek, M.; Asada, R.; Bauernfeind, D.; Baumgartner, R. Prospective sustainability assessment: The case of wood in automotive applications. *Int. J. Life Cycle Assess.* 2020, 25, 2027–2049. [CrossRef]
- Ladenika, A.; Bodunrin, M.; Burman, N.; Croft, J.; Engelbrecht, S.; Goga, T.; MacGregor, O.; Maepa, M.; Harding, K. Assessing the availability of life cycle assessments in Austria. *Int. J. Life Cycle Assess.* 2019, 24, 614–619. [CrossRef]
- 144. ISO 14067; Greenhouse Gases—Carbon Footprint of Products—Requirements and Guidelines for Quantification. ISO: Geneva, Switzerland, 2018.
- Sphera. GaBi Databases 2022 Edition. 2022. Available online: https://sphera.com/wp-content/uploads/2022/02/Database_ Upgrade_2022_Upgrades_and_improvements_Wo_clean.pdf (accessed on 21 August 2023).
- 146. Hydro. Low-Carbon and Recycled Aluminium. 2022. Available online: https://www.hydro.com/en-AT/aluminium/products/low-carbon-and-recycled-aluminium (accessed on 21 August 2023).
- 147. BMK. Österreichs Infoportal zur Energiesituation. 2023. Available online: https://energie.gv.at/ (accessed on 21 August 2023). (In German)
- 148. European Environment Agency. Greenhouse Gas Emission Intensity of Electricity Generation. 2023. Available online: https://www.eea.europa.eu/ds_resolveuid/ef6559ba07b14a3f9aff7602e61bedd4 (accessed on 21 August 2023).

- 149. Statista. Major Car Companies' Five-Year Average Net Profit Margin. 2022. Available online: https://www.statista.com/ statistics/1186661/car-company-profit-margin/ (accessed on 21 August 2023).
- 150. Wellbrock, W.; Ludin, D.; Röhrle, L.; Gerstlberger, W. Sustainability in the automotive industry, importance of and impact on automobile interior—Insights from an empirical survey. *Int. J. Corp. Soc. Responsib.* **2020**, *5*, 10. [CrossRef]
- 151. Hannon, E.; Nauclér, T.; Suneson, A.; Yüksel, F. The Zero-Carbon Car: Abating Material Emissions Is Next on the Agenda. 2020. Available online: https://www.mckinsey.com/capabilities/sustainability/our-insights/the-zero-carbon-car-abating-materialemissions-is-next-on-the-agenda (accessed on 21 August 2023).
- 152. Horton, P.; Allwood, J.; Cassell, P.; Edwards, C.; Tautscher, A. Material demand reduction and closed-loop recycling automotive aluminium. *MRS Adv.* **2018**, *3*, 1393–1398. [CrossRef]
- 153. Muslemani, H.; Liang, X.; Kaesehage, K.; Ascui, F.; Wilson, J. Opportunities and challenges for decarbonizing steel production by creating markets for 'green steel' products. J. Clean. Prod. 2021, 315, 128127. [CrossRef]
- 154. International Aluminium Institute. Global Aluminium Cycle. 2021. Available online: https://alucycle.international-aluminium. org/public-access/ (accessed on 21 August 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.